

GOPEN ACCESS

Citation: Sherling ES, Perrin AJ, Knuepfer E, Russell MRG, Collinson LM, Miller LH, et al. (2019) The *Plasmodium falciparum* rhoptry bulb protein RAMA plays an essential role in rhoptry neck morphogenesis and host red blood cell invasion. PLoS Pathog 15(9): e1008049. https://doi.org/ 10.1371/journal.ppat.1008049

Editor: Oliver Billker, Umea Universitet, SWEDEN

Received: May 27, 2019

Accepted: August 27, 2019

Published: September 6, 2019

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.ppat.1008049

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

RESEARCH ARTICLE

The *Plasmodium falciparum* rhoptry bulb protein RAMA plays an essential role in rhoptry neck morphogenesis and host red blood cell invasion

Emma S. Sherling¹, Abigail J. Perrin¹, Ellen Knuepfer², Matthew R. G. Russell³, Lucy M. Collinson³, Louis H. Miller⁴, Michael J. Blackman^{1,5}*

1 Malaria Biochemistry Laboratory, The Francis Crick Institute, London, United Kingdom, 2 Malaria Parasitology Laboratory, The Francis Crick Institute, London, United Kingdom, 3 Electron Microscopy Science Technology Platform, The Francis Crick Institute, London, United Kingdom, 4 Laboratory of Malaria and Vector Research, National Institute of Allergy and Infectious Diseases, National Institutes of Health, Rockville, Maryland, United States of America, 5 Faculty of Infectious and Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom

* mike.blackman@crick.ac.uk

Abstract

The malaria parasite Plasmodium falciparum invades, replicates within and destroys red blood cells in an asexual blood stage life cycle that is responsible for clinical disease and crucial for parasite propagation. Invasive malaria merozoites possess a characteristic apical complex of secretory organelles that are discharged in a tightly controlled and highly regulated order during merozoite egress and host cell invasion. The most prominent of these organelles, the rhoptries, are twinned, club-shaped structures with a body or bulb region that tapers to a narrow neck as it meets the apical prominence of the merozoite. Different protein populations localise to the rhoptry bulb and neck, but the function of many of these proteins and how they are spatially segregated within the rhoptries is unknown. Using conditional disruption of the gene encoding the only known glycolipid-anchored malarial rhoptry bulb protein, rhoptry-associated membrane antigen (RAMA), we demonstrate that RAMA is indispensable for blood stage parasite survival. Contrary to previous suggestions, RAMA is not required for trafficking of all rhoptry bulb proteins. Instead, RAMA-null parasites display selective mislocalisation of a subset of rhoptry bulb and neck proteins (RONs) and produce dysmorphic rhoptries that lack a distinct neck region. The mutant parasites undergo normal intracellular development and egress but display a fatal defect in invasion and do not induce echinocytosis in target red blood cells. Our results indicate that distinct pathways regulate biogenesis of the two main rhoptry sub-compartments in the malaria parasite.

Author summary

Despite improved control measures over recent decades, malaria is still a considerable health burden across much of the globe. The disease is caused by a single-celled parasite

PLOS | PATHOGENS

Funding: ESS was in receipt of a Wellcome Trust/ National Institutes of Health PhD studentship (103459/Z/14/Z), whilst AJP was supported by Wellcome Trust grant 106239/Z/14/A (to MJB). Work in the LHM laboratory was supported by the Intramural Research Program of the Division of Intramural Research, National Institute of Allergy and Infectious Diseases, NIH. This work was supported by funding to MJB from the Francis Crick Institute (https://www.crick.ac.uk/) which receives its core funding from Cancer Research UK (FC001043; https://www.cancerresearchuk.org), the UK Medical Research Council (FC001043; https://www.mrc.ac.uk/), and the Wellcome Trust (FC001043; https://wellcome.ac.uk/). The work was also supported by Wellcome Trust ISSF2 funding to the London School of Hygiene & Tropical Medicine. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

that invades and replicates within host cells. During invasion, the parasite discharges a set of flask-shaped secretory organelles called rhoptries, the contents of which are crucial for invasion as well as for modifications to the host cell that are important for parasite survival. Rhoptry discharge occurs through fusion of the relatively elongated rhoptry neck to the apical surface of the parasite. Different proteins reside within the bulbous rhoptry body and the neck regions, but how these proteins are selectively sent to their correct subcompartments within the rhoptries and how the rhoptries are formed, is poorly understood. Here we show that a malaria parasite rhoptry bulb protein called rhoptry-associated membrane antigen (RAMA) plays an essential role in rhoptry neck formation and correct trafficking of certain rhoptry neck and bulb proteins. Parasites deficient in RAMA produce malformed rhoptries and–probably as a result—cannot invade host red blood cells. Our work sheds new light on how rhoptries are formed and reveals insights into the mechanism by which the correct sorting of proteins to distinct regions of the rhoptry is regulated.

Introduction

Malaria is a devastating disease of tropical and subtropical regions. Requiring a mammalian host and a mosquito vector for transmission, at least six species of the genus Plasmodium cause disease in humans, with *Plasmodium falciparum* being responsible for the great majority of mortality. All the manifestations of clinical disease result from repeated cycles of invasion, replication within and lytic egress from red blood cells (RBC). Invasion is an orchestrated process, comprising several steps including merozoite attachment, deformation of the RBC membrane, merozoite reorientation, formation of a high affinity interaction between the apical zone of the merozoite and the RBC surface, active entry, and finally sealing of the RBC membrane behind the intracellular parasite [1-3]. Invasion is generally immediately followed by a period of transient RBC echinocytosis, a morphological transformation of the RBC surface into an undulated or 'spiky' appearance, although this can also be induced under certain conditions even in the absence of successful invasion [3–5]. Entry into the host cell occurs concomitantly with formation of a membrane-bound parasitophorous vacuole (PV) within which the invading parasite comes to rest. The parasite then transforms within minutes into a 'ring' stage form before initiating intracellular development, progressing through a mononuclear trophozoite stage to a multinucleated schizont which undergoes segmentation to form a new generation of daughter merozoites. Parasite-induced rupture of the PV membrane (PVM) and host RBC membrane eventually enables egress of the merozoites to initiate a fresh erythrocytic cvcle.

Egress and invasion involve the regulated discharge of at least four classes of secretory organelles-rhoptries, micronemes, exonemes and dense granules—that are unique to apicomplexan parasites and that are positioned within the apical end of the merozoite [6]. The largest of these organelles, the rhoptries, are twinned club or pear-shaped structures that are formed *de novo* during each blood-stage life cycle by progressive fusion of post-Golgi vesicles, with the possible involvement of an endosome-like pathway [7–9]. Rhoptries are characterised by a relatively wide bulb region which narrows to a neck or duct towards the apical prominence of the merozoite; rhoptry discharge takes place via the duct. Despite the bulb and neck regions not being separated by a discernible membranous boundary, examination by transmission electron microscopy (TEM) has shown a distinct sub-compartmentalisation of each rhoptry, with the bulb and neck regions displaying different staining characteristics in both *Plasmodium* [7]

and the related apicomplexan parasite Toxoplasma gondii [10]. Based on the detailed examination of the fate of a subset of rhoptry neck proteins (RONs; [11]) and bulb proteins, it has been suggested that the contents of the different rhoptry sub-compartments are released at different points in the invasion pathway, with the RONs being discharged before rhoptry bulb proteins [12]. Consistent with this, several RON proteins have been extensively implicated in the early stages of host cell entry, notably in formation of the moving junction or tight junction (TJ), a short-lived doughnut-shaped structure through which the parasite passes as it enters the host cell (e.g. [13–17]) Some members of the *P. falciparum* reticulocyte binding protein homologue (RH) family of RBC adhesins, which also appear to localise to both the rhoptry bulb and neck [18], also play central roles in early steps in invasion (e.g. [3, 19, 20]). However, the association between function, timing of discharge and compartmentalisation within the rhoptries may not be a strict one, since the *Plasmodium*-specific protein RON12 is predominantly released postinvasion, into the nascent PV [21]. Similarly, a rhoptry bub location does not preclude roles during invasion, since the rhoptry bulb protein RhopH3, a component of the high molecular weight (HMW) RhopH complex, is important for RBC entry [22]. In contrast, RhopH3 and its partner proteins RhopH1/Clag and RhopH2 have additionally been shown to be involved in nutrient uptake during development of the intracellular parasite [22–25], whilst the rhoptry bulb low molecular weight (LMW) RAP1/RAP2 complex has been implicated in PVM formation [26]. In *Toxoplasma* (but not in *Plasmodium*) several rhoptry bulb proteins are enzymes, including proteases, phosphatases and kinases or pseudokinases, the latter two groups of which dramatically modulate host cell STAT signalling and the immunity-related GTPase (IRG) pathway involved in controlling parasite replication, and play key roles in virulence [27, 28]. Collectively, the current evidence points to multiple diverse roles for rhoptry proteins, with a general theme being that rhoptry neck proteins are often but not exclusively involved in host cell entry whilst rhoptry bulb proteins usually fulfil subsequent roles in the life cycle (see [29-31] for excellent reviews of this subject). Despite these many insights, of the ~60 known and putative *Plasmodium* rhoptry proteins identified through a combination of numerous antibody studies and proteomic analyses of isolated apicomplexan rhoptries [11, 32-34], the molecular functions of very few Plasmodium rhoptry proteins have been characterised. Crucially, how the different subclasses of *Plasmodium* rhoptry proteins are selectively delivered to their distinct sub-compartments within the organelle is unknown.

Rhoptry-associated membrane antigen (RAMA) was initially identified in P. falciparum as a rhoptry bulb-resident protein which is expressed relatively early in the asexual blood stage cycle, before the de novo formation of nascent rhoptries [35, 36]. Like many other Plasmodium rhoptry bulb proteins, RAMA has no obvious orthologue in Toxoplasma or other coccidian Apicomplexa. Unique amongst known Plasmodium rhoptry proteins, RAMA is associated with the inner (lumenal) face of the rhoptry bulb membrane via a glycosyl phosphatidylinositol (GPI) membrane anchor [36]. Trafficking of RAMA to the rhoptries shares several features of other rhoptry proteins, including transport via the Golgi apparatus and sensitivity to brefeldin A. The ~170 kDa RAMA precursor then undergoes proteolytic processing in the rhoptries or nascent rhoptries, involving removal of an N-terminal segment to generate a mature ~60 kDa form (p60). To gain insights into its function, Topolska and colleagues [36] and Richard et al. (2009) [37] used fluorescent resonance energy transfer (FRET) and immunoprecipitation to demonstrate that RAMA appears to interact with both RhopH3 and RAP1, proteins which are enriched in lipid rafts but that lack lipid anchors. The authors hypothesised that RAMA plays an escorter role in recruiting and trafficking these proteins to the developing rhoptries, and that proteolytic cleavage of RAMA then facilitates their dissociation from the escorter complex. RAMA p60 (but not the RAMA precursor) was detectable in free extracellular merozoites and evidence was also presented that the protein is released from rhoptries at invasion,

supported by the observation that a recombinant form of the extreme C-terminal region of RAMA could bind the RBC membrane. However, in newly invaded 'ring' stage parasites, RAMA p60 was found associated with the PVM, indicating that at least some of the protein was transported into the host cell with the invading parasite [36]. Very recent work has suggested that in fact the RAMA-RAP1 complex may itself be cargo for the Plasmodium orthologue of sortilin, an integral membrane protein that in mammalian cells acts as an escorter to transport proteins to the lysosomes, endocytic pathway and plasma membrane [38]. This is consistent with a role in rhoptry formation, since in Toxoplasma sortilin is required for biogenesis of both rhoptries and a second class of secretory organelle called micronemes [39], and the P. falciparum sortilin orthologue has been localised to the Golgi [9]. Although restricted to the Plasmodium genus, RAMA orthologues were identified in all Plasmodium species examined [40], and this was subsequently confirmed by the now extensive genome sequence data from numerous additional Plasmodium species (see PlasmoDB RAMA gene ID: PF3D7 070 7300; https://plasmodb.org/plasmo/). More recent evidence from targeted or global reverse genetics studies indicates that disruption of the P. falciparum or P. berghei RAMA gene is deleterious or lethal [41-43], consistent with an important role for RAMA in the asexual blood stage parasite life cycle. However, the molecular function of RAMA has remained obscure.

Here we have used a robust conditional mutagenesis approach to examine the function of RAMA in the *P. falciparum* asexual blood stage life cycle. In contrast to previous suggestions, we found that disruption of RAMA expression does not affect trafficking of the rhoptry bulb proteins RhopH3 and RAP1, nor biogenesis of the rhoptry bulb. Instead, our results indicate a selective role for RAMA in formation of the rhoptry neck structure. Perhaps as a result, the RAMA mutants show mislocalisation of several RON proteins and a lethal defect in host RBC invasion.

Results

Efficient conditional disruption of the P. falciparum RAMA gene

Previous attempts to directly disrupt the P. falciparum RAMA gene using conventional targeted genetic techniques were unsuccessful [41], suggesting an important role for RAMA in the haploid asexual blood stages. To gain insights into the molecular function of RAMA we therefore used a conditional gene disruption approach based on the rapamycin-inducible DiCre conditional recombinase system. To enable this, we first modified the RAMA gene in the DiCre-expressing 1G5DC P. falciparum line [44] to introduce 'silent' loxP motifs suitable to act as sites for DiCre-mediated site-specific recombination. The RAMA gene encodes an 861 amino acid residue protein and comprises 4 small exons plus a single much larger exon (exon 2). The complexity and relatively large size (~3.2 kb) of the RAMA locus precluded facile floxing of the entire coding sequence in a single manipulation as previously performed with other *P. falciparum* genes [4], so we took an alternative approach in which we inserted into exon 2 a synthetic construct combining two heterologous loxPint elements (i.e. synthetic introns each containing a loxP site; [45]), flanking 176 bp of intervening recodonised exon sequence. The inserted sequence was precisely integrated in a marker-free manner in a single Cas9-promoted homologous recombination step (Fig 1A). The resulting modified RAMA gene (referred to as *RAMAloxP*) thus effectively comprised 7 exons with an internal recodonised exon flanked by loxPint sequences. DiCre-mediated recombination between the introduced *loxP* sites was predicted to excise this exon. Importantly, excision was also expected to create a frame-shift in the coding sequence immediately downstream of the single chimeric loxPint site remaining after excision, leading to a severely truncated RAMA coding sequence encoding only the N-terminal 220 residues, referred to below as RAMAAE2.



Fig 1. Conditional disruption of the P. falciparum RAMA gene. (A) The RAMA gene (PlasmoDB PF3D7_0707300) comprises 5 exons (numbered grey boxes; not to scale) and 4 introns (blue lines). Two AG-AT motifs (splice site junction consensus sequences; [72] in exon 2, separated by 176 bp, were chosen as the location for insertion of two loxP-containing (purple arrowhead) heterologous introns (SERA2:loxPint, blue line; SUB2:loxPint, pink line) flanking a recodonised form of the intervening 176 bp coding sequence (E2 recod, green box). The integration sequence, provided on repair construct pESS_RAMA_E2INT_loxP, was inserted by double crossover homologous recombination between homology arms (E2H1 and E2H2), facilitated using Cas9 and sgRNAs encoded on a separate plasmid to generate a targeted double stranded break (see Materials and methods). DiCre-mediated recombination between the loxP sites in the modified RAMAloxP locus was expected to excise the E2 recod sequence, leaving only a single chimeric SERA2:loxP:SUB2 intron on the chromosome and generating a frame-shift following the first 220 amino acid codons of the RAMA coding sequence. Positions of primers used for diagnostic PCR are indicated (half arrows; see S1 Table for numerical codes and sequences of all primers used in this study), as are XmnI restriction sites (X) used for Southern blot analysis. The grey dotted line indicates the region used for hybridisation of a probe for Southern blot. (B) PCR analysis of RAMAloxP clones 9C10 and 12C7 confirmed the expected gene modification events. Amplicons diagnostic of the wild type locus obtained from parental 1G5DC parasites (wild type; wt) were not amplified from the RAMAloxP clones. Primer identities are colour-coded as in panel (A). (C) Southern blot; XmnI-digested genomic DNA from RAMAloxP clones 9C10 and 12C7 was hybridised with a probe specific for the E2H2 sequence. Expected fragment sizes are 2,191 bp (wt) and 1,050 bp (RAMAloxP). (D) Diagnostic PCR showing efficient RAP-induced excision of the floxed sequence in the RAMAloxP parasite clones, leading in both cases to the expected decrease in amplicon size from 594 bp to 304 bp. Primer identities are colour-coded as in panel (A). (E) Immunoblots of SDS PAGE-fractionated cycle 0 schizont extracts probed with anti-RAMA-D antibodies (raised against residues 482-758 of the RAMA sequence; [36]). Loss of RAMA p60 expression (arrowed) was almost complete in the RAP-treated RAMAloxP parasites. Similarly-treated 1G5DC schizonts (wt) were analysed in parallel to confirm that RAP-treatment has no effect on RAMA expression in the absence of the RAMAloxP modification. The blots were stripped and re-probed with anti-BiP antibodies as a loading control (lower panels). (F) Representative IFA images demonstrating loss of RAMA expression upon RAP-treatment of RAMAloxP-9C10 parasites. Cycle 0 schizonts were co-stained with anti-RAMA-D antibodies and the anti-RAP1 monoclonal antibody (mAb) 4F3. 4,6-diamidino-2-phenylindole (DAPI) was used as a DNA stain. The great majority of RAPtreated parasites (>99.8% as calculated from exhaustive examination of 1,002 schizonts by IFA) showed no detectable anti-RAMA-D staining, but the RAP1 signal remained unaffected. Scale bars, 5 µm.

https://doi.org/10.1371/journal.ppat.1008049.g001

Limiting dilution cloning of the transfected parasite population resulted in the isolation of 2 parasite clones called *RAMAloxP*-9C10 and *RAMAloxP*-12C7, each independently generated using different sgRNAs. The expected modification of the native *RAMA* locus was confirmed in both parasite clones by diagnostic PCR (Fig 1B) and Southern blot (Fig 1C) and the clones were therefore used for all subsequent experiments. Growth assays comparing replication rates of *RAMAloxP*-9C10 and *RAMAloxP*-12C7 parasites with the parental 1G5DC line revealed no significant differences, indicating that the modifications made to the *RAMA* gene did not detectably impair parasite growth (S1 Fig).

To examine the efficiency of conditional excision of the floxed *RAMA* sequence, tightly synchronised newly-invaded ring stage cultures of both the *RAMAloxP* clones were divided into two and treated for just 4 h with either rapamycin (RAP; 100 nM final) or DMSO vehicle control (1% v/v, mock-treated). Following washing and further incubation for ~44 h to allow maturation to mature schizont stage, parasite genomic DNA (gDNA) was extracted and examined by analytical PCR. This revealed highly efficient RAP-induced excision of the floxed *RAMA* sequence (Fig 1D). To determine the impact of excision on RAMA expression, the schizonts were analysed using a rabbit polyclonal antibody called anti-RAMA-D, specific for a C-terminal segment (amino acid residues 482 to 758) of RAMA [36]. As shown in Fig 1E and Fig 1F, little or no signal was detected by the anti-RAMA-D antibodies in schizonts of the RAP-treated cultures, either by Western blot or by indirect immunofluorescence analysis (IFA). These results confirmed the PCR data, demonstrating in two independent *RAMAloxP* parasite clones essentially complete conditional disruption of RAMA expression within a single erythrocytic cycle.

RAMA is required for P. falciparum replication in vitro

The IFA images in Fig 1F indicated that, despite efficient disruption of the RAMA gene in RAP-treated RAMAloxP parasites, the parasites were able to form multinucleated schizonts towards the end of the erythrocytic growth cycle in which they were RAP-treated (henceforth referred to as cycle 0). To examine the effects of gene disruption on longer-term parasite viability, the mock- and RAP-treated RAMAloxP clones were maintained in culture and parasite replication monitored by flow cytometry over the ensuing erythrocytic cycles. Mock- and RAP-treated parental 1G5DC parasites were also included in these experiments to control for any off-target effects of RAP treatment. As shown in Fig 2A, a dramatic reduction in parasite replication rate was quickly evident in the RAP-treated RAMAloxP parasites, indicating a defect in parasite proliferation. This was confirmed by diagnostic PCR (Fig 2B) which revealed that upon passage of the cultures for up 9 erythrocytic cycles, the excised locus quickly became undetectable in RAP-treated cultures (indicating disappearance of the RAMAAE2 mutants) whilst in contrast the initially very minor fraction of non-excised parasites gradually expanded to take over the cultures. This was consistent with an acute fitness defect in the RAMA Δ E2 mutants. A further quantitative assessment of the impact of RAMA disruption upon parasite viability was obtained by comparing the capacity of mock- and RAP-treated RAMAloxP parasites to form zones of erythrocyte lysis, or plaques [46], in static cultures in flat-bottomed microplates (Fig 2C). This showed that RAP-treatment of both the RAMAloxP-9C10 and RAMAloxP-12C7 clones produced a highly significant reduction in plaque formation; the mean relative plaque forming capacity (RPFC) of RAP-treated parasites was $3.39\% \pm 0.833$ (meaning that for every 10,000 plaques produced by the control mock-treated cultures, only 339 were formed in the same number of RAP-treated culture-containing wells) (p = 3.32 x 10^{-8} , t = 115.9, d.f. = 4). Together, these data confirmed that RAMA plays a critical role in *P*. falciparum asexual blood stage viability.



Fig 2. RAMA is required for asexual blood stage *P. falciparum* **replication**. (A) Replication of *RAMAloxP* and parental 1G5DC parasites over the course of 4–5 erythrocytic cycles following mock-treatment (DMSO) or RAP-treatment of synchronous ring-stage cultures. Parasitaemia values (quantified by flow cytometry) were averaged from three biological replicate experiments performed using blood from different donors, and are presented as mean \pm SD. Posthoc comparisons by three-way ANOVA revealed significant differences between growth curves of the mock- and RAP-treated *RAMAloxP* clones 9C10 (p = 0.00995) and 12C7 (p = 0.00860), but not of the parental 1G5DC line (p = 0.998). (B) Diagnostic PCR analysis of mock- or RAP-treated cultures of *RAMAloxP* clone 9C10 parasites, indicating gradual overgrowth of the excised population by an initially undetectable minority of non-excised parasites. Parasite gDNA sampled at each cycle over the course of 9 erythrocytic growth cycles was analysed (see Fig 1A for primer identities). The signal specific for the excised locus became less intense with time whilst the signal specific for the non-excised locus increased in intensity, eventually becoming the sole PCR amplicon detectable. (C) Plots comparing plaque formation by mock- and RAP-treated cultures of *RAMAloxP* clones 9C10 and 12C7. Cultures were dispensed into flat-bottomed microwell plates at equivalent parasite densities and plaques scored following static incubation for 14 days. Error bars represent \pm 1 SD from two independent experiments in blood from different donors. A total of 180 wells were used for each treatment in each replicate of the assay.

https://doi.org/10.1371/journal.ppat.1008049.g002

Disruption of RAMA produces an invasion defect

Given the established role of rhoptries in invasion, we considered that the replication defect shown by the RAMA Δ E2 parasites might most likely be explained by a reduced capacity to invade host cells. To investigate this, mature *RAMAloxP*-9C10 and *RAMAloxP*-12C7 cycle 0 schizonts were isolated from highly synchronous mock- or RAP-treated cultures then added to fresh RBCs to achieve a similar starting parasitaemia. Following further incubation for 4 h to allow schizont rupture, the cultures were stained with Hoechst 33342 and the proportions of newly-invaded ring-stage parasites determined by flow cytometry. This consistently showed that the RAP-treated parasites displayed substantially reduced ring formation, corresponding to only ~10% of that of mock-treated parasites (Fig 3A).

As egress and invasion are intimately linked, we explored whether the lack of ring formation in these assays was indicative of a defect in egress from the host erythrocyte. To do this the capacity of RAMA Δ E2 parasites to undergo egress was examined by time-lapse differential interference contrast (DIC) microscopy. As shown in Fig 3B, no detectable differences in the efficiency, kinetics or morphology of merozoite egress from RAP-treated *RAMAloxP* parasites was evident compared to their mock-treated counterparts. It was concluded that disruption of RAMA does not affect egress, but results in a severe defect in productive invasion of new host erythrocytes.

To characterize this invasion defect in further detail, the behaviour of naturally-released merozoites and their interactions with host RBCs was examined by time-lapse DIC video microscopy. As shown in S1 Movie and S2 Movie, and quantified in Fig 3C, merozoites released from RAP-treated (RAMA Δ E2) cycle 0 schizonts interacted initially with neighbouring host RBCs in a similar manner to their mock-treated *RAMAloxP* counterparts, inducing repeated and often strong deformation of the host cells. However, unlike the mock-treated merozoites, in no case (out of a total of 20 similar egress events examined) did we observe successful invasion by the RAMA Δ E2 merozoites. Of particular additional interest, no RBC echinocytosis was ever observed in the case of the RAMA Δ E2 merozoite-RBC interactions. Echinocytosis generally takes place shortly following invasion (see S1 Movie), but under certain conditions can also occur in the absence of invasion, where it is thought to indicate discharge of rhoptry components [3–5]. It was concluded that loss of RAMA selectively results in a fatal block in invasion, likely due to a defect in rhoptry discharge.

Disruption of RAMA leads to mislocalisation of some but not all rhoptry neck and rhoptry bulb proteins

As mentioned above, morphological characterisation of mature cycle 0 RAMA Δ E2 schizonts by light microscopy (IFA) indicated apparently normal schizont development (Fig 1F). However, given the clear invasion phenotype displayed by the RAMA Δ E2 mutants and the known rhoptry bulb localisation of RAMA [36], we decided to examine the structure of the mutant parasites in greater detail to assess the possible impact of RAMA disruption on the makeup and morphology of the rhoptries. This was considered particularly important given previous claims that RAMA interacts with RhopH3 and RAP1 and plays a role in the trafficking of both proteins to the rhoptries [36, 37].

To determine the effects of RAMA disruption on the expression and trafficking of other rhoptry proteins, mature cycle 0 schizonts from mock-treated and RAP-treated *RAMAloxP* cultures were probed by IFA using a suite of antibodies specific for a range of rhoptry bulb proteins. As shown in <u>Fig 4A</u>, this revealed that the subcellular localisation and staining intensity of the LMW rhoptry complex protein RAP2, the HMW rhoptry complex proteins RhopH1/Clag3.1, RhopH2 and RhopH3, the invasion ligand Rh5, and the rhoptry-associated



Fig 3. Disruption of RAMA results in an invasion defect. (A) Reduced ring formation by RAMA mutants. Mature cycle 0 schizonts isolated from mock- or RAP-treated cultures of the indicated clones were incubated with fresh RBCs and ring-stage parasitaemia determined 4 h later by flow cytometry. Data were averaged from three biological replicate experiments, using blood

from different donors. Error bars, ± 1 SD. Statistical significance was determined by a two-tailed unpaired t-test where $p \le 0.0001$, ***** and p > 0.05, non-significant (ns). Ring formation by the parental 1G5DC parasites was unaffected by RAP treatment (t = 0.507, d.f. = 4, p = 0.639). (B) Still images from experiments monitoring egress of merozoites by time-lapse DIC microscopy of mature mock- and RAP-treated *RAMAloxP* clone 9C10 schizonts. Highly synchronous cycle 0 schizonts were incubated with the parasite cGMP-dependent protein kinase G inhibitor (4-[7-[(dimethylamino)methyl]-2-(4-fluorphenyl)imidazo[1,2- α]pyridine-3-yl]pyrimidin-2-amine (compound 2; C2) for 4 h to reversibly arrest egress [73]. Immediately after washing away the C2 the parasites were monitored for 30 minutes. No delay in egress was observed in the RAP-treated population, and there were no discernible differences in egress efficiency or morphology. Time in minutes from commencement of microscopy is indicated. Scale bar, 10 µm. Images shown are from clone 9C10 but are representative of 3 independent replicate experiments examining *RAMAloxP* clones 9C10 and 12C7. (C) Quantification of invasion, merozoite-induced RBC surface deformation and echinocytosis observed by video microscopy following rupture of mock- or RAP-treated *RAMAloxP* clone 9C10 schizonts. At least 20 videos per condition were quantified. Statistical significance was assessed by t-test; ns indicates not significant (p > 0.05) whereas **** indicates p < 0.0001. The RAP-treated (RAMAAE2) merozoites interacted with and deformed target RBCs, but neither invaded them nor induced echinocytosis. See also S1 Movie and S2 Movie.

https://doi.org/10.1371/journal.ppat.1008049.g003

protein ARO (thought to localise to the cytosolic face of the rhoptry bulb and neck [47–49]) were unaffected by RAMA disruption; all retained their characteristic apical punctate rhoptry signal in the RAP-treated *RAMAloxP* schizonts. Note that similar findings were noted above for RAP1 (Fig 1F). It was concluded that truncation of RAMA does not discernibly affect the trafficking of a number of established rhoptry bulb proteins.

To examine the effects of RAMA truncation upon the subcellular localisation of RON proteins, mock- and RAP-treated RAMAloxP schizonts were next probed with antibodies specific for RON2 [50], RON3, or RON4 [51]. As shown in Fig 4B and Fig 4C, in all cases this revealed evidence for RON protein mislocalisation in the RAMAAE2 mutants, although the precise phenotype appeared to depend upon the extent of parasite maturity. Specifically, for all three RON proteins, in immature schizonts a much reduced signal intensity was detectable which lacked the punctate, apically-disposed pattern typical of rhoptries. In highly mature segmented schizonts, IFA signals for RON2, RON3 and RON4 were often completely absent. These findings were particularly intriguing since, whilst RON2 and RON4 are established rhoptry neck proteins that play roles in TJ formation, RON3 has been localised by immuno electron microscopy to the rhoptry bulb in *Plasmodium* [52], rather than to the rhoptry neck as originally determined for the *Toxoplasma* orthologue [11]. Work in *Toxoplasma* also supports a rhoptry bulb location (see ToxoDB annotation for the TgRON3 gene at https://toxodb.org/toxo/app/ record/gene/TGME49_223920). The current nomenclature for RON3 is therefore somewhat misleading, and these results therefore suggested that disruption of RAMA expression can modulate the localisation of rhoptry neck proteins as well as at least one rhoptry bulb protein.

To investigate this issue further using additional rhoptry markers, we examined *RAMAloxP* schizonts by IFA using antibodies to two proteins that have been previously localised to either the rhoptry neck (RON12; [21]) or both the rhoptry body and the neck (Rh2b; [18]) in *P. falciparum* merozoites. As shown in S2 Fig and in contrast to the effects on RON2, RON3 and RON4, disruption of RAMA expression had no discernible effect on expression or localisation of either RON12 or Rh2b.

As expected, disruption of RAMA did not impact on the localisation of the major merozoite surface protein MSP1 or the microneme protein apical membrane antigen 1 (AMA1) (Fig 4D). AMA1 is often observed to translocate onto the merozoite surface in highly mature schizonts, and this phenomenon too was observed at similar frequency in both mock-treated and RAP-treated *RAMAloxP* schizonts (Fig 4D, lower panels). Collectively, these results indicated that disruption of RAMA results in selective mislocalisation and/or loss of rhoptry neck and certain rhoptry bulb proteins, whilst having no detectable impact on the expression and trafficking of other rhoptry proteins examined.



Fig 4. Disruption of RAMA leads to selective mislocalisation of some but not all rhoptry proteins. (A) IFA showing that the staining profile of RAP2, RhopH1/Clag3.1, Rh5, RhopH2, ARO and RhopH3 was unaltered between DMSO- and RAP-treated *RAMAloxP* clone 9C10 schizonts. Scale bar, 2 μm. (B) IFA showing that, whilst RhopH3 and RhopH1/Clag3.1 staining was unaffected by RAMA disruption, RON2, RON3 and RON4 staining was modified in RAP-treated *RAMAloxP* clone 9C10 schizonts. In all cases, the staining was much weaker in the RAMAΔE2 parasites and lacked the characteristically punctate,

apically-disposed pattern typical of rhoptry staining. Scale bar, 2 μ m. (C) The extent of RON protein mis-trafficking depends on the maturity of the RAPtreated *RAMAloxP* schizonts. Whilst some signal was still detectable in less mature schizonts (those with fewer nuclei), the RON2, RON3 and RON4-specific signal was absent or much reduced in the mature segmented schizonts (examples indicated with arrows). Note that these images are of lower magnification that those shown in the rest of the figure in order to display multiple schizonts in each field. Scale bar, 10 μ m. (D) Schizonts of DMSO- and RAP-treated *RAMAloxP* clone 9C10 were probed with a polyclonal antibody against AMA1 as well as anti-MSP1 antibodies. AMA1 localisation (upper panel) and discharge onto the merozoite surface in mature schizonts (lower panel) was unaffected by RAMA disruption. Scale bar, 2 μ m. Parasite nuclei were visualised throughout by staining with DAPI.

https://doi.org/10.1371/journal.ppat.1008049.g004

RAMA is selectively required for formation of the rhoptry neck

Previous studies of RAMA suggested a role in rhoptry biogenesis [36]. Our discovery that some rhoptry proteins become mislocalised upon RAMA disruption prompted us to next examine the impact of RAMA disruption on rhoptry formation and morphology. To do this, mature segmented cycle 0 mock- or RAP-treated *RAMAloxP* schizonts were examined by TEM. To quantify rhoptry formation, the number of rhoptry profiles detectable in individual intracellular merozoites in 70 nm thin sections was recorded. This revealed no statistically significant differences in the total number of visible rhoptry profiles between the control and RAMA Δ E2 parasites (glm fit with Poisson model, predicting rhoptry count from status, nonzero status coefficient test p-value = 0.232) (Fig 5A). This result was in complete accord with the IFA results obtained with the rhoptry bulb-specific antibodies described in Fig 4, suggesting that the overall number of rhoptry organelles assembled in wild-type and RAMA Δ E2 parasites was similar.

During the above analysis, we noticed that whilst other intracellular features of the control and RAP-treated parasites (including microneme morphology) were similar, there was a relative absence of typical club-shaped rhoptry profiles in the RAMA Δ E2 TEM micrographs. To attempt to quantify these apparent differences in rhoptry morphology, the TEM rhoptry images were categorised as being either circular or club-shaped based on blinded visual examination (Fig 5B). As shown in Fig 5C, whilst rhoptries of both shapes were observed in merozoites of mock-treated and RAP-treated RAMAloxP parasites, a significantly lower proportion of club-shaped profiles was observed in the RAMA Δ E2 samples. Moreover, in those RAMA Δ E2 rhoptries that visibly possessed an apical projection, these appeared qualitatively different from wild type, being relatively short and stubby compared to the elongated projections typical of the latter (Fig 5B). This difference in shape was confirmed by quantifying the aspect ratio (the ratio of the longest to shortest diameter) in the electron micrographs of all non-circular rhoptries from both control and RAP-treated parasites. The resulting data (Fig 5D) clearly indicated that the length of the rhoptry neck projections in control parasites (mean aspect ratio = 1.73) was significantly greater than those of RAP-treated parasites (mean aspect ratio = 1.21) (p < 0.0001, t = 5.29, d.f. = 58). Since these profiles represent random section planes through rhoptries, it was considered unlikely that the differences noted were a chance result of random variation in the section plane obtained during sample preparation. However, to finally examine this point, electron tomography was performed on serial 200 nm thick sections of a single mock-treated RAMAloxP schizont and a RAP-treated RAMAloxP schizont (S3 Movie and S4 Movie). This allowed computational three-dimensional modelling of much of the rhoptry content of each schizont by segmentation and rendering of the tomograms. The resulting models (Fig 5E) supported the thin section TEM data in showing a clear difference between the elongated shape of rhoptries in the DMSO-treated RAMAloxP schizonts and the strikingly more spherical nature of the RAMA Δ E2 rhoptries.

Collectively, our EM data indicated that disruption of RAMA does not affect the number of rhoptries formed, but has a major impact on their morphology, leading to the generation of



Fig 5. Disruption of RAMA leads to defective rhoptry morphology and reduced rhoptry neck formation. Tightly synchronised mature schizonts from DMSO- or RAP-treated cultures of *RAMAloxP* clone 9C10 parasites were fixed and processed for TEM, then three sections from each sample taken and 20

schizonts examined per section. (A) Mean number of visible rhoptry profiles per schizont calculated from each set of 60 TEM images of DMSO-treated and RAP-treated schizonts. There was no statistically significant difference in the mean number of rhoptry profiles visible. Error bars, ± 1 SD. p>0.05, non-significant (ns). (B) Representative TEM images from DMSO and RAP-treated *RAMAloxP* schizonts indicating that club-shaped and circular (i.e. likely spherical) rhoptry profiles were detected in both. Scale bar, 100 nm. (C) The proportion of club-shaped rhoptry profiles visible in DMSO-treated *RAMAloxP* schizonts was greater than that in RAP-treated *RAMAloxP* schizonts. Statistical significance was determined by unpaired t-test. Error bars depict ± 1 SD. p ≤ 0.01 , **. (D) Plots of aspect ratio (longest diameter divided by shortest diameter) calculated from thin section electron micrographs for all rhoptries (both DMSO and RAP-treated *RAMAloxP* schizonts. Statistical difference determined by two-tailed t-test. Error bars, ± 1 SD. p ≤ 0.001 , ***. (E) Two dimensional cross section images and three-dimensional models derived from 200 nm thick section serial tomograms acquired from a DMSO-treated (upper two panels) and a RAP-treated *RAMAloxP* schizont (lower two panels). Rhoptries are depicted in various colours to easily distinguish individual structures. Whilst the rhoptries of the DMSO-treated schizonts appear predominantly club shaped, those of the RAP-treated parasites appear much more circular with stuby, non-conoid projections. Scale bars, 500 nm. All observations where zero rhoptry profiles were reported for a merozoite were deemed 'failed' observations for the purposes of this analysis and were not included in the analysis.

https://doi.org/10.1371/journal.ppat.1008049.g005

relatively spherical or 'neckless' rhoptries. This may explain why certain rhoptry neck proteins mislocalise in the RAMA mutants.

Discussion

The involvement of rhoptry proteins in host cell entry has long been proposed, supported by early observations that rhoptry discharge coincides temporally with invasion in several apicomplexan parasites, including *Toxoplasma* [53–56] and *P. falciparum* [17]. Those studies have been burgeoned by recent demonstrations that correct positioning of rhoptries at the apical pole of the *Toxoplasma* tachyzoite is a prerequisite for invasion (but not egress) [48] whilst in both *P. falciparum* merozoites and *Toxoplasma* tachyzoites rhoptries appear to undergo fusion with each other during invasion [30, 57]. The discovery of the role of several RON proteins in TJ formation [15, 16] was consistent with this overall model for rhoptry function, but as discussed in the Introduction it is also now clear that rhoptry proteins play roles following completion of invasion in such diverse functions as PVM generation, subversion of host cell signalling, and nutrient acquisition. Here we have established a new role for a *Plasmodium* rhoptry protein in biogenesis of these important organelles.

We draw three major conclusions from our study. First, we have shown that RAMA is essential for parasite survival, and that disruption of RAMA expression results in a selective, fatal defect in host cell invasion with no obvious effect on intracellular parasite replication or egress. This finding is reminiscent of the results of conditional ablation in Toxoplasma of ARO, a palmitoylated protein localised to the cytoplasmic rhoptry surface which is essential for correct tethering of the rhoptries within the tachyzoite apex [47-49]. Loss of TgARO results in rhoptries that are dispersed throughout the cytoplasm of the parasite, and the mutant tachyzoites undergo normal egress but cannot invade host cells. As well as not being able to penetrate targeted RBCs, we found that RAMA Δ E2 merozoites showed no capacity to induce echinocytosis in interacting RBCs. This provides important insights into the underlying defect. Evidence that rhoptry discharge can take place upon contact with the host cell even in the absence of productive invasion was first demonstrated through observations of the appearance of small nascent intraerythrocytic channels or vacuoles at the site of apical attachment of P. knowlesi merozoites to the RBC surface; this occurred both under normal, invasion-permissive conditions but also in cultures containing cytochalasin B, an inhibitor of actin polymerisation that blocks invasion downstream of TJ formation [58–60]. Subsequent work in P. falciparum combining electron microscopy with super-resolution immunofluorescence microscopy showed that these vacuoles contain rhoptry proteins [13]. Similar structures (termed 'evacuoles' when formed in the presence of cytochalasin) were observed upon interaction of Toxoplasma tachyzoites with host cells, and these were also found to contain large amounts of rhoptry bulb proteins [53, 61]. More recent comprehensive microscopic examination of the

interactions between free *P. falciparum* merozoites and host RBCs provided evidence that rhoptry discharge is the primary inducer of echinocytosis in target RBCs [3], and consistent with the above observations this can occur even in the absence of productive invasion. Very recent work has added further support to this model, showing that merozoites genetically deficient in key components of the cyclic nucleotide signalling pathway or the molecular motor that drives invasion can induce echinocytosis in target RBCs, despite being unable to penetrate the cells [4, 5]. It has thus become clear that host RBC echinocytosis—irrespective of whether it is followed by successful invasion—provides a useful reporter for rhoptry function in *Plasmodium*. In our current study, the complete absence of echinocytosis upon interactions between RAMAAE2 merozoites and target RBCs is therefore completely consistent with a defect in rhoptry discharge and/or the absence of the (currently unidentified) rhoptry component(s) that induce echinocytosis. RAMA was previously proposed to act as an invasion ligand based on indications that anti-RAMA antibodies inhibit merozoite invasion of RBCs [36] and that recombinant RAMA-based peptides could apparently bind to RBCs and inhibit invasion [62]. Our study did not directly address this proposed function for RAMA.

The second clear conclusion from our study is that RAMA is not required for correct trafficking of RAP1 or RhopH3, as previously proposed based on protein-protein interaction and co-localisation studies [36–38], nor for trafficking of several other rhoptry bulb proteins examined. Our results are therefore not consistent with a rhoptry bulb-specific protein escorter role for RAMA. However, our findings may not necessarily be in conflict with the interaction model; RAMA may associate with RhopH3 and RAP1, for example, in order to facilitate formation of the respective HMW and LMW complexes, whilst other factors may regulate localisation of these protein complexes to the rhoptry bulb. It is important to note that our RAMA disruption strategy did not remove the entire gene and could in principle result in the expression of the N-terminal 220 residues of the protein as a truncated protein product (Fig 1). We did not have the antibody tools to detect whether such a truncated protein would lack the C-terminal sequences between residues 315–840 previously shown to interact with RAP1 and sortilin, and required for trafficking to the rhoptries [36–38]. We are therefore confident that our strategy generated a loss-of-function mutant, and the phenotype supports that.

Third, our results show that RAMA is instead required for correct localisation of a subset of RON and rhoptry bulb proteins and for formation of the rhoptry neck, providing the first genetic evidence that RAMA plays an essential role in rhoptry biogenesis. The causal relationship between these phenotypes is unclear, in that we cannot discriminate between whether a defective rhoptry morphology leads to mislocalisation of the RON proteins or alternatively whether correct trafficking of RON proteins is required for rhoptry neck morphogenesis. RAMA may interact with RON proteins to ensure their correct trafficking, although no associations between RAMA and RON proteins were identified in the earlier studies [36, 37]. The involvement of a rhoptry bulb protein in trafficking of rhoptry neck proteins appears somewhat counterintuitive. Could RAMA direct these proteins to the bulb then regulate their onward trafficking to the nascent neck domain? It is interesting to note that RON12, a rhoptry neck protein localisation of which was unaffected by RAMA disruption in this study, has very recently been confirmed as a rhoptry neck protein in merozoites of the rodent malaria parasite Plasmodium berghei, but a rhoptry bulb protein in P. berghei sporozoites, a developmental stage of the parasite generated in the mosquito vector [63]. This suggests that the trafficking characteristics of rhoptry proteins may not be an intrinsic function of their structure *per se*, but dependent upon the developmental context of their expression.

Both RON2 and RON4 are key for formation of the TJ, so their mis-trafficking may explain in part the invasion defect in the RAMAΔE2 mutants. However, we propose that the most

dramatic element of the phenotype-the defect in rhoptry neck formation-is most likely primarily responsible for the loss of invasive capacity in the mutants. Absence of the neck structure would presumably prevent docking and fusion of the rhoptries with the parasite plasma membrane at the merozoite apical prominence, reducing or ablating the capacity for rhoptry discharge. It might be expected that loss of the rhoptry neck structure would also result in a phenotype similar to that of loss of TgARO [48, 49], with the rhoptries becoming dispersed throughout the merozoite cytoplasm rather than apically located; however the relatively small dimensions of the *P. falciparum* merozoite relative to the *Toxoplasma* tachyzoite likely limits cytoplasmic movement of untethered rhoptries, rendering this phenotype less obvious.

In conclusion, the selective impact of RAMA disruption on the correct localisation of some but not all rhoptry proteins supports the existence of subsets of rhoptry proteins, some of which require RAMA for correct trafficking whilst others do not. We have shown that RAMA disruption is lethal, likely through a direct impact upon rhoptry biogenesis and a resulting indirect impact upon invasion but not egress. Given that there is no obvious orthologue of RAMA in *Toxoplasma* and other apicomplexan parasites outside of the *Plasmodium* genus, our findings raise obvious questions as to how rhoptry neck formation is regulated in other Apicomplexa. In this regard is interesting to note that disruption of the gene encoding a GPIanchored *Toxoplasma* rhoptry bulb protein called TgCA_RP results in dysmorphic, fragmented rhoptries [64], although the phenotype is quite different from that of the RAMAΔE2 *P. falciparum* mutants and the morphologies of rhoptries are also distinct between the two apicomplexan genera. TgCA_RP and RAMA share no homology, but it is conceivable that the roles of the two proteins in rhoptry biogenesis may be similar.

Materials and methods

Ethics statement

This work used human blood that was provided anonymously through the UK Blood Transfusion Service.

Reagents and antibodies

Oligonucleotide primers (see <u>S1 Table</u>) were purchased from Sigma-Aldrich (Gillingham, UK). Rapamycin was also obtained from Sigma-Aldrich and stored frozen as a stock solution (10 µM) in DMSO. The antifolate drug WR99210 (Jacobus Pharmaceuticals, Princeton, NJ) was stored as a 20 µM stock solution in DMSO. The anti-RAMA-D, anti-RhopH3 and anti-KAHRP antibodies were kind gifts of Ross Coppel (Monash University, Australia) whilst Osamu Kaneko (Nagasaki University, Japan) kindly provided antibodies against RhopH1/ Clag3.1. The polyclonal anti-AMA1 antibody has previously been described [65], as has the RhopH2-specific mAb 61.3 [66]. The anti-MSP1 mAb 89.1, the anti-RON3 mAb 1H1, the anti-RAP1 mAb 4F3, a rabbit anti-ARO polyclonal antibody, and rabbit antibodies against RON12 were kind gifts of Tony Holder (Francis Crick Institute, UK), and the antibodies to Rh2b were generously provided by Julian Rayner (University of Oxford, UK (anti-MAHRP1 antibody), Simon Draper, University of Oxford, UK (anti-MAHRP1 antibody), Takafumi Tsuboi, Ehime University Japan (anti-RON2 antibody) and Jean-François Dubremetz, Université Montpellier 2, France (anti-RON4 mAb 24C6).

Generation of RAMA conditional mutant parasite lines

RAMAloxP mutant *P. falciparum* lines were generated by transfecting parasites with 60 µg of a repair plasmid alongside 20 µg of a plasmid for expression of Cas9. Repair plasmid pESS_RA-MA_E2INT_loxP (synthesised by ThermoFisher Scientific) comprised synthetic heterologous *SERA2:loxP* and *SUB2:loxP* introns which were inserted at AG-AT sites 776 bp and 952 bp respectively downstream of the *RAMA* gene ATG start codon. The intervening 174 bp was codon-optimised for expression in *Escherichia coli*. Sequence comprising 568 bp of exon 2 both 5' and 3' to the site of the heterologous intron insertions was included to act as flanking regions for homologous recombination. EuPaGDT software was used to identify 20 bp protospacer sequences (flanked by a 5'-NGG-3' protospacer adjacent motif) in order to specifically target Cas9-mediated cleavage to the *RAMA* gene locus. pRAMAsgRNA1 and pRAMAsgRN A2 were created by modifying the Cas9-sgRNA-encoding pDC2 vector (a kind gift of Marcus Lee, Sanger Institute, UK) by introduction of sgRNAs 1 and 2 generated by annealing complementary primer sequences RAMA_sgRNA_E2_F to RAMA_sgRNA_E2_R (sgRNA1), and RAMA_sgRNA_E2S_F to RAMA_sgRNA_E2S_R (sgRNA2).

P. falciparum cell culture, transfection and cloning

All parasite work described used the 3D7-derived DiCre-expressing 1G5DiCre (1G5DC) *P. falciparum* clone [44]. Parasites were maintained in culture and synchronised using standard procedures [67, 68]. DNA constructs were introduced by electroporation into mature schizont-stage 1G5DC parasites using previously described methods [69]. Approximately 24 h post-transfection the cultures were supplemented with WR99210 (2.5 nM) to allow selection of parasites carrying the human *dhfr* selectable marker encoded by the pRAMAsgRNA plasmids. Parasites were maintained under drug selection for 2 erythrocytic cycles before being transferred to drug-free culture medium. Following confirmation of integration in the resulting parasite populations by diagnostic PCR, parasites were cloned by limiting dilution [46]. Conditional excision of floxed genomic sequences in the transgenic parasite clones was induced as previously described [44] by treating tightly synchronized ring stage parasite cultures with 100 nM RAP in 1% (v/v) DMSO (or the same concentration of DMSO only as a vehicle control) for 4 h at 37°C.

Detection of integration and excision events by diagnostic PCR, Southern blotting and immunoblotting

5' and 3' integration of pESS_RAMA_E2INT_loxP into the genome was detected by PCR using the integration specific primers RAMA_exon1_F plus RAMA_exon2recod_R, and RAMA_exon2recod_F plus RAMA_exon2_R. No amplification was expected from the wild type specific primer set RAMA_exon1_F plus RAMA_exon2WT_R. Diagnostic PCR was also used to confirm that the integrated DiCre expression cassette remained unmodified in isolated clones, using primers SERA5_DiCre_F and DiCre_integrated. Reversion to the wild type *SERA5* locus by spontaneous excision of the integrated DiCre cassette (which can occur at low levels in the 1G5DC line; [4]) was indicated by the amplification of a 1700 bp amplicon using primers specific to the unmodified *SERA5* locus (SERA5_DiCre_F and SERA5_DiCre_R). To further confirm the genotype of potential *RAMAloxP* integrant clones, a 400 bp digoxigenin (DIG)-labelled probe corresponding to the 5' 400 bp of the region of exon 2 used as the second homology arm was generated by PCR amplification using primers RAMA_SB_homology2_F and RAMA_SB_homology2_R, performed in the presence of DIG-labelled nucleotides provided by the PCR DIG Labelling Mix Plus (Roche). Approximately 5 µg of gDNA from

putative integrant cultures was digested with an excess of XmnI restriction enzyme, then fractionated by agarose gel electrophoresis, transferred overnight to Amersham Hybond N+ nylon membranes (GE Healthcare Life Sciences), and cross-linked with UV light (1200 µjoules/cm²). After pre-hybridisation, hybridisation, washing and blocking steps, hybridised signal was detected by incubation with polyclonal alkaline phosphate-conjugated anti-DIG antibodies (Roche). The membrane was incubated with detection buffer and CDP-star (Roche), before being exposed to X-ray film. RAP-induced recombination between genomic *loxP* sites was detected by PCR analysis of schizont-stage gDNA (harvested 42–44 h following mock- or RAP-treatment) using primers RAMA_Harm1_F and RAMA_Harm2_R. For immunoblot analysis, SDS-PAGE fractionated proteins were transferred to nitrocellulose membranes before being blocked, incubated with primary antibody diluted to the appropriate concentration, washed and then probed with a HRP-conjugated secondary antibody as previously described [65]. After final washes, membranes were incubated with Immobilon Western Chemiluminescent HRP Substrate (Merck) and signals visualised by exposure to X-ray film.

Indirect immunofluorescence assays

Thin blood films of mock- or RAP-treated Percoll-enriched RAMAloxP schizonts were airdried and stored desiccated at -80°C. As required, samples were thawed at 37°C and parasites fixed in 4% (w/v) paraformaldehyde and permeabilised in 0.1% (v/v) Triton X-100. Fixed slides were blocked then probed with the relevant antibody diluted as follows: rabbit anti-RAMA-D, 1:1000; mouse mAb 4F3 (anti-RAP1), 1:100; rabbit anti-RhopH1/Clag3.1, 1:100; mouse mAb MRA876 (anti-RAP2), 1:200; rabbit anti-Rh5, 1:10,000; mouse mAb 61.3 (anti-RhopH2), 1:100; rabbit anti-ARO, 1:500; rabbit anti-RhopH3 (anti-Ag44), 1:2000; rabbit anti-RON2, 1:250; mouse mAb 1H1 (anti-RON3), 1:100; mouse mAb 24C6 (anti-RON4), 1:500; rabbit anti-AMA1, 1:500; mouse mAb 89.1 (anti-MSP1), 1:1000; rabbit anti-RON12, 1:5000; rabbit anti-Rh2b, 1:500. After incubation and washing, slides were probed with the required Alexa Fluor 488 or 594-conjugated secondary antibody diluted 1:10,000. Slides were then mounted in ProLong Gold Antifade Mountant containing DAPI (ThermoFisher Scientific), sealed with Cytoseal-60 (Thermofisher Scientific) and images collected using a Nikon Eclipse Ni-E wide field microscope with a Hamamatsu C11440 digital camera and 100x/1.45NA oil immersion objective. Identical exposure conditions were used at each wavelength for each pair of mock- and RAP-treated samples under comparison. Images were processed using Fiji software.

Parasite growth and invasion assays

To determine the growth capability of mutant parasites over multiple cycles, DMSO and RAPtreated cultures were adjusted to a parasitaemia of 0.1%, with samples for flow cytometry being sampled for fixation every 48 h for up to 6 erythrocytic growth cycles. Pelleted parasites were fixed in 4% paraformaldehyde (Electron Microscopy Sciences) and 0.02% glutaraldehyde (Sigma-Aldrich) in PBS for 1 h at 37°C, before being diluted into PBS. When ready for flow cytometry, samples were stained with a 1:10,000 dilution of Hoechst 33342 (ThermoFisher Scientific) in PBS for 30 min at 37°C. The invasive capacity of parasites was determined by adding RBCs to highly synchronous Percoll-purified mature schizonts to obtain a parasitaemia of 1%. After further incubation for 4 h, cultures were fixed and the percentage of newly infected RBCs was determined. Parasitaemia and DNA replication were indicated by Hoechst-staining intensity determined using a LSRFortessa (BD Biosciences) flow cytometer, collecting a minimum of 10⁵ events per sample. Data were analysed using FlowJo software. The relative plaqueforming capacity of parasites was determined as previously described [46, 70]. All experiments were performed in triplicate using blood from different donors, and statistical analysis was performed using GraphPad Prism.

Time-lapse video microscopy

To visualise merozoite egress, schizont-stage mock- or RAP-treated parasites were imaged as previously described [4, 69]. Invasion videos were similarly performed using schizonts purified from mock- or RAP- treated cultures mixed with fresh uninfected RBCs. DIC images were acquired every 150 ms and Nikon NIS Elements AR analysis software was used to produce the resulting time-lapse videos.

Transmission electron microscopy

Schizonts of *RAMAloxP* parasites were harvested ~44 h following mock- or RAP-treatment, before being fixed in 2.5% glutaraldehyde and 4% formaldehyde in 0.1 M phosphate buffer. After fixation, cells were washed in 0.1 M phosphate buffer, embedded in 2% agarose and processed as described previously [4].

Statistical analysis of TEM images

For each sample, three sections were taken from each of three of the 1 mm³ blocks of agaroseembedded schizonts. Twenty schizonts from each block were quantified, scoring the number of rhoptries within each intracellular merozoite. Each rhoptry was also visually classified as being circular or club-shaped (i.e. possessing distinct neck and bulb regions). Statistical analyses of the data were performed using R 3.3.1 (https://www.R-project.org/) (R Core Team, 2013). Generalised linear models were fit using the glm() function, with the first model (rhoptry_count~DMSO_or_Rapa_status, family = Poisson) being used to test for a difference in rhoptry counts. For comparison of 'club-shaped' counts on the non-zero rhoptry data, the model used was club_count~DMSO_or_Rapa_status + rhoptry_count, family = Poisson. Analysis of Deviance was performed using the anova() function on the resultant glm object, with test = 'Chisq'.

Transmission electron tomography

Tilt series images of a schizont from each experimental condition were collected from +60° to -60° with 1° increments using the Tecnai User Interface software (FEI). The 1024 x 1024 pixel images were recorded with an Ultrascan charge coupled device camera (Gatan Inc.) with a pixel size of 5.42 nm. The IMOD package [71] was used to reconstruct individual tomograms, with patch tracking used to create a fiducial model. For each schizont, tomograms from three serial 200 nm sections were flattened and joined together in z to obtain a continuous volume. The 3dmod programme of IMOD was used to manually segment rhoptry membranes, from which three-dimensional surface models were generated. Models were left open where rhoptries extended outside the joined volume, and where the top or bottom surface of a rhoptry was lost between adjacent tomograms.

Supporting information

S1 Fig. No significant differences in replication rates of *RAMAloxP P. falciparum* clones and parental 1G5DC parasites over the course of 4 erythrocytic cycles. Parasitaemia values (quantified by flow cytometry) were averaged from three biological replicate experiments performed using blood from different donors, and are presented as mean \pm SD. (TIF) **S2 Fig. Disruption of RAMA has no effect on subcellular localisation of the rhoptry neck protein RON12 or the rhoptry body and neck protein Rh2b.** IFA showing that the staining profiles of both RON12 and Rh2b were unaltered between DMSO- and RAP-treated *RAMA-loxP* clone 9C10 schizonts. Antibodies to RhopH2 were used as a marker for the rhoptry bulb. Scale bar, 5 µm.

(TIF)

S1 Table. Oligonucleotide primer sequences used in this study. (PDF)

S1 Movie. Time-lapse video microscopy showing that mock-treated (DMSO-treated) *RAMAloxP* clone 9C10 merozoites bind to and deform RBCs, invade and induce echinocytosis.

(MP4)

S2 Movie. Time-lapse video microscopy showing that RAP-treated, RAMA-deficient *RAMAloxP* clone 9C10 merozoites interact with and deform host RBCs but neither invade nor induce echinocytosis. (MP4)

S3 Movie. Tomogram video and three-dimensional reconstruction of rhoptries from a mock-treated (DMSO-treated) *RAMAloxP* clone 9C10 schizont (see <u>Materials and methods</u> for a description of methods used). (MOV)

S4 Movie. Tomogram video and three-dimensional reconstruction of rhoptries from of a RAP-treated *RAMAloxP* clone 9C10 schizont (see <u>Materials and methods</u> for a description of methods used). (MOV)

Acknowledgments

The authors would like to extend special thanks to Ross Coppel for extensive discussions and the generous gift of anti-RAMA-D antibodies. They are also indebted to Fiona Hackett for invaluable support in generating and maintaining the DiCre-expressing parent lines, and thank Judith Green and Tony Holder for the generous donation of antibody reagents.

Author Contributions

Conceptualization: Emma S. Sherling, Louis H. Miller, Michael J. Blackman.

Data curation: Matthew R. G. Russell.

Formal analysis: Emma S. Sherling, Abigail J. Perrin, Ellen Knuepfer, Matthew R. G. Russell, Lucy M. Collinson.

Funding acquisition: Emma S. Sherling, Louis H. Miller, Michael J. Blackman.

Investigation: Emma S. Sherling, Abigail J. Perrin, Ellen Knuepfer, Matthew R. G. Russell.

Methodology: Abigail J. Perrin, Ellen Knuepfer, Matthew R. G. Russell, Lucy M. Collinson.

Supervision: Lucy M. Collinson, Louis H. Miller, Michael J. Blackman.

Validation: Abigail J. Perrin.

Visualization: Abigail J. Perrin.

Writing - original draft: Emma S. Sherling, Abigail J. Perrin, Michael J. Blackman.

Writing – review & editing: Emma S. Sherling, Ellen Knuepfer, Louis H. Miller, Michael J. Blackman.

References

- 1. Gilson PR, Crabb BS. Morphology and kinetics of the three distinct phases of red blood cell invasion by Plasmodium falciparum merozoites. Int J Parasitol. 2009; 39(1):91–6. https://doi.org/10.1016/j.ijpara. 2008.09.007 PMID: 18952091.
- Dvorak JA, Miller LH, Whitehouse WC, Shiroishi T. Invasion of erythrocytes by malaria merozoites. Science. 1975; 187(4178):748–50. PMID: 803712.
- Weiss GE, Gilson PR, Taechalertpaisarn T, Tham WH, de Jong NW, Harvey KL, et al. Revealing the sequence and resulting cellular morphology of receptor-ligand interactions during Plasmodium falciparum invasion of erythrocytes. PLoS Pathog. 2015; 11(2):e1004670. Epub 2015/02/28. https://doi.org/ 10.1371/journal.ppat.1004670 PMID: 25723550; PubMed Central PMCID: PMC4344246.
- Perrin AJ, Collins CR, Russell MRG, Collinson LM, Baker DA, Blackman MJ. The Actinomyosin Motor Drives Malaria Parasite Red Blood Cell Invasion but Not Egress. MBio. 2018; 9(4). Epub 2018/07/05. https://doi.org/10.1128/mBio.00905-18 PMID: 29970464; PubMed Central PMCID: PMC6030552.
- Patel A, Perrin AJ, Flynn HR, Bisson C, Withers-Martinez C, Treeck M, et al. Cyclic AMP signalling controls key components of malaria parasite host cell invasion machinery. PLoS Biol. 2019; 17(5): e3000264. Epub 2019/05/11. https://doi.org/10.1371/journal.pbio.3000264 PMID: 31075098.
- Cowman AF, Tonkin CJ, Tham WH, Duraisingh MT. The Molecular Basis of Erythrocyte Invasion by Malaria Parasites. Cell Host Microbe. 2017; 22(2):232–45. Epub 2017/08/12. <u>https://doi.org/10.1016/j.chom.2017.07.003</u> PMID: 28799908.
- Bannister LH, Hopkins JM, Fowler RE, Krishna S, Mitchell GH. Ultrastructure of rhoptry development in Plasmodium falciparum erythrocytic schizonts. Parasitology. 2000; 121 (Pt 3):273–87. Epub 2000/11/ 21. PMID: 11085247.
- Schrevel J, Asfaux-Foucher G, Hopkins JM, Robert V, Bourgouin C, Prensier G, et al. Vesicle trafficking during sporozoite development in Plasmodium berghei: ultrastructural evidence for a novel trafficking mechanism. Parasitology. 2008; 135(Pt 1):1–12. Epub 2007/10/03. <u>https://doi.org/10.1017/</u> S0031182007003629 PMID: 17908361.
- Krai P, Dalal S, Klemba M. Evidence for a Golgi-to-endosome protein sorting pathway in Plasmodium falciparum. PloS one. 2014; 9(2):e89771. Epub 2014/03/04. https://doi.org/10.1371/journal.pone. 0089771 PMID: 24587025; PubMed Central PMCID: PMC3934947.
- Lemgruber L, Lupetti P, De Souza W, Vommaro RC. New details on the fine structure of the rhoptry of Toxoplasma gondii. Microsc Res Tech. 2011; 74(9):812–8. Epub 2011/09/01. <u>https://doi.org/10.1002/jemt.20960</u> PMID: 23939668.
- Bradley PJ, Ward C, Cheng SJ, Alexander DL, Coller S, Coombs GH, et al. Proteomic analysis of rhoptry organelles reveals many novel constituents for host-parasite interactions in Toxoplasma gondii. J Biol Chem. 2005; 280(40):34245–58. Epub 2005/07/09. <u>https://doi.org/10.1074/jbc.M504158200</u> PMID: 16002398.
- Zuccala ES, Gout AM, Dekiwadia C, Marapana DS, Angrisano F, Turnbull L, et al. Subcompartmentalisation of proteins in the rhoptries correlates with ordered events of erythrocyte invasion by the blood stage malaria parasite. PloS one. 2012; 7(9):e46160. Epub 2012/10/11. <u>https://doi.org/10.1371/journal.pone.0046160</u> PMID: 23049965; PubMed Central PMCID: PMC3458004.
- Riglar DT, Richard D, Wilson DW, Boyle MJ, Dekiwadia C, Turnbull L, et al. Super-resolution dissection of coordinated events during malaria parasite invasion of the human erythrocyte. Cell Host Microbe. 2011; 9(1):9–20. Epub 2011/01/18. doi: S1931-3128(10)00413-0 [pii] https://doi.org/10.1016/j.chom. 2010.12.003 PMID: 21238943.
- Lamarque M, Besteiro S, Papoin J, Roques M, Vulliez-Le Normand B, Morlon-Guyot J, et al. The RON2-AMA1 interaction is a critical step in moving junction-dependent invasion by apicomplexan parasites. PLoS Pathog. 2011; 7(2):e1001276. Epub 2011/02/25. https://doi.org/10.1371/journal.ppat. 1001276 PMID: 21347343; PubMed Central PMCID: PMC3037350.
- Alexander DL, Mital J, Ward GE, Bradley P, Boothroyd JC. Identification of the moving junction complex of Toxoplasma gondii: a collaboration between distinct secretory organelles. PLoS Pathog. 2005; 1(2): e17. Epub 2005/10/26. https://doi.org/10.1371/journal.ppat.0010017 PMID: 16244709; PubMed Central PMCID: PMC1262624.

- Lebrun M, Michelin A, El Hajj H, Poncet J, Bradley PJ, Vial H, et al. The rhoptry neck protein RON4 relocalizes at the moving junction during Toxoplasma gondii invasion. Cell Microbiol. 2005; 7(12):1823– 33. Epub 2005/11/29. https://doi.org/10.1111/j.1462-5822.2005.00646.x PMID: 16309467.
- Aikawa M, Miller LH, Johnson J, Rabbege J. Erythrocyte entry by malarial parasites. A moving junction between erythrocyte and parasite. J Cell Biol. 1978; 77(1):72–82. Epub 1978/04/01. https://doi.org/10. 1083/jcb.77.1.72 PMID: 96121; PubMed Central PMCID: PMC2110026.
- Duraisingh MT, Triglia T, Ralph SA, Rayner JC, Barnwell JW, McFadden GI, et al. Phenotypic variation of Plasmodium falciparum merozoite proteins directs receptor targeting for invasion of human erythrocytes. EMBO J. 2003; 22(5):1047–57. Epub 2003/02/28. https://doi.org/10.1093/emboj/cdg096 PMID: 12606570; PubMed Central PMCID: PMC150330.
- Baum J, Chen L, Healer J, Lopaticki S, Boyle M, Triglia T, et al. Reticulocyte-binding protein homologue 5—an essential adhesin involved in invasion of human erythrocytes by Plasmodium falciparum. Int J Parasitol. 2009; 39(3):371–80. Epub 2008/11/13. https://doi.org/10.1016/j.ijpara.2008.10.006 PMID: 19000690.
- Rodriguez M, Lustigman S, Montero E, Oksov Y, Lobo CA. PfRH5: a novel reticulocyte-binding family homolog of plasmodium falciparum that binds to the erythrocyte, and an investigation of its receptor. PloS one. 2008; 3(10):e3300. Epub 2008/10/02. https://doi.org/10.1371/journal.pone.0003300 PMID: 18827878; PubMed Central PMCID: PMC2553180.
- Knuepfer E, Suleyman O, Dluzewski AR, Straschil U, O'Keeffe AH, Ogun SA, et al. RON12, a novel Plasmodium-specific rhoptry neck protein important for parasite proliferation. Cell Microbiol. 2014; 16 (5):657–72. Epub 2013/08/14. https://doi.org/10.1111/cmi.12181 PMID: 23937520; PubMed Central PMCID: PMC3922828.
- Sherling ES, Knuepfer E, Brzostowski JA, Miller LH, Blackman MJ, van Ooij C. The Plasmodium falciparum rhoptry protein RhopH3 plays essential roles in host cell invasion and nutrient uptake. Elife. 2017;6. Epub 2017/03/03. <u>https://doi.org/10.7554/eLife.23239</u> PMID: <u>28252384</u>; PubMed Central PMCID: PMC5365315.
- Counihan NA, Chisholm SA, Bullen HE, Srivastava A, Sanders PR, Jonsdottir TK, et al. Plasmodium falciparum parasites deploy RhopH2 into the host erythrocyte to obtain nutrients, grow and replicate. Elife. 2017; 6. Epub 2017/03/03. <u>https://doi.org/10.7554/eLife.23217</u> PMID: <u>28252383</u>; PubMed Central PMCID: PMC5365316.
- Ito D, Schureck MA, Desai SA. An essential dual-function complex mediates erythrocyte invasion and channel-mediated nutrient uptake in malaria parasites. Elife. 2017; 6. Epub 2017/02/22. https://doi.org/ 10.7554/eLife.23485 PMID: 28221136; PubMed Central PMCID: PMC5349850.
- Gupta A, Balabaskaran-Nina P, Nguitragool W, Saggu GS, Schureck MA, Desai SA. CLAG3 Self-Associates in Malaria Parasites and Quantitatively Determines Nutrient Uptake Channels at the Host Membrane. MBio. 2018; 9(3). Epub 2018/05/10. https://doi.org/10.1128/mBio.02293-17 PMID: 29739907; PubMed Central PMCID: PMC5941077.
- Ghosh S, Kennedy K, Sanders P, Matthews K, Ralph SA, Counihan NA, et al. The Plasmodium rhoptry associated protein complex is important for parasitophorous vacuole membrane structure and intraerythrocytic parasite growth. Cell Microbiol. 2017; 19(8). Epub 2017/02/17. <u>https://doi.org/10.1111/cmi.</u> 12733 PMID: 28205319.
- Boothroyd JC. Have it your way: how polymorphic, injected kinases and pseudokinases enable Toxoplasma to subvert host defenses. PLoS Pathog. 2013; 9(4):e1003296. Epub 2013/05/02. https://doi. org/10.1371/journal.ppat.1003296 PMID: 23633947; PubMed Central PMCID: PMC3635977.
- Behnke MS, Dubey JP, Sibley LD. Genetic Mapping of Pathogenesis Determinants in Toxoplasma gondii. Annu Rev Microbiol. 2016; 70:63–81. Epub 2016/07/01. https://doi.org/10.1146/annurev-micro-091014-104353 PMID: 27359216.
- Kats LM, Black CG, Proellocks NI, Coppel RL. Plasmodium rhoptries: how things went pear-shaped. Trends in parasitology. 2006; 22(6):269–76. Epub 2006/04/26. https://doi.org/10.1016/j.pt.2006.04.001 PMID: 16635585.
- Dubremetz JF. Rhoptries are major players in Toxoplasma gondii invasion and host cell interaction. Cell Microbiol. 2007; 9(4):841–8. Epub 2007/03/10. <u>https://doi.org/10.1111/j.1462-5822.2007.00909.x</u> PMID: 17346309.
- Counihan NA, Kalanon M, Coppel RL, de Koning-Ward TF. Plasmodium rhoptry proteins: why order is important. Trends in parasitology. 2013; 29(5):228–36. Epub 2013/04/11. https://doi.org/10.1016/j.pt. 2013.03.003 PMID: 23570755.
- Oakes RD, Kurian D, Bromley E, Ward C, Lal K, Blake DP, et al. The rhoptry proteome of Eimeria tenella sporozoites. Int J Parasitol. 2013; 43(2):181–8. Epub 2012/12/25. https://doi.org/10.1016/j. ijpara.2012.10.024 PMID: 23262303.

- Sam-Yellowe TY, Banks TL, Fujioka H, Drazba JA, Yadav SP. Plasmodium yoelii: novel rhoptry proteins identified within the body of merozoite rhoptries in rodent Plasmodium malaria. Exp Parasitol. 2008; 120(1):113–7. Epub 2008/07/09. https://doi.org/10.1016/j.exppara.2008.05.001 PMID: 18606406.
- Sam-Yellowe TY, Florens L, Wang T, Raine JD, Carucci DJ, Sinden R, et al. Proteome analysis of rhoptry-enriched fractions isolated from Plasmodium merozoites. J Proteome Res. 2004; 3(5):995–1001. Epub 2004/10/12. https://doi.org/10.1021/pr049926m PMID: 15473688.
- Smythe JA, Coppel RL, Brown GV, Ramasamy R, Kemp DJ, Anders RF. Identification of two integral membrane proteins of Plasmodium falciparum. Proc Natl Acad Sci U S A. 1988; 85(14):5195–9. Epub 1988/07/01. https://doi.org/10.1073/pnas.85.14.5195 PMID: 3293051; PubMed Central PMCID: PMC281715.
- Topolska AE, Lidgett A, Truman D, Fujioka H, Coppel RL. Characterization of a membrane-associated rhoptry protein of Plasmodium falciparum. J Biol Chem. 2004; 279(6):4648–56. Epub 2003/11/14. https://doi.org/10.1074/jbc.M307859200 PMID: 14613941.
- Richard D, Kats LM, Langer C, Black CG, Mitri K, Boddey JA, et al. Identification of rhoptry trafficking determinants and evidence for a novel sorting mechanism in the malaria parasite Plasmodium falciparum. PLoS Pathog. 2009; 5(3):e1000328. Epub 2009/03/07. https://doi.org/10.1371/journal.ppat. 1000328 PMID: 19266084; PubMed Central PMCID: PMC2648313.
- Hallee S, Counihan NA, Matthews K, de Koning-Ward TF, Richard D. The malaria parasite Plasmodium falciparum Sortilin is essential for merozoite formation and apical complex biogenesis. Cell Microbiol. 2018; 20(8):e12844. Epub 2018/03/28. https://doi.org/10.1111/cmi.12844 PMID: 29582546.
- Sloves PJ, Delhaye S, Mouveaux T, Werkmeister E, Slomianny C, Hovasse A, et al. Toxoplasma sortilin-like receptor regulates protein transport and is essential for apical secretory organelle biogenesis and host infection. Cell Host Microbe. 2012; 11(5):515–27. Epub 2012/05/23. https://doi.org/10.1016/j. chom.2012.03.006 PMID: 22607804.
- Topolska AE, Black CG, Coppel RL. Identification and characterisation of RAMA homologues in rodent, simian and human malaria species. Mol Biochem Parasitol. 2004; 138(2):237–41. Epub 2004/11/24. https://doi.org/10.1016/j.molbiopara.2004.05.018 PMID: 15555735.
- Sanders PR, Kats LM, Drew DR, O'Donnell RA, O'Neill M, Maier AG, et al. A set of glycosylphosphatidyl inositol-anchored membrane proteins of Plasmodium falciparum is refractory to genetic deletion. Infect Immun. 2006; 74(7):4330–8. Epub 2006/06/23. <u>https://doi.org/10.1128/IAI.00054-06</u> PMID: <u>16790807</u>; PubMed Central PMCID: PMC1489731.
- Zhang M, Wang C, Otto TD, Oberstaller J, Liao X, Adapa SR, et al. Uncovering the essential genes of the human malaria parasite Plasmodium falciparum by saturation mutagenesis. Science. 2018; 360 (6388). Epub 2018/05/05. https://doi.org/10.1126/science.aap7847 PMID: 29724925; PubMed Central PMCID: PMC6360947.
- Bushell E, Gomes AR, Sanderson T, Anar B, Girling G, Herd C, et al. Functional Profiling of a Plasmodium Genome Reveals an Abundance of Essential Genes. Cell. 2017; 170(2):260–72 e8. Epub 2017/ 07/15. https://doi.org/10.1016/j.cell.2017.06.030 PMID: 28708996; PubMed Central PMCID: PMC5509546.
- 44. Collins CR, Das S, Wong EH, Andenmatten N, Stallmach R, Hackett F, et al. Robust inducible Cre recombinase activity in the human malaria parasite Plasmodium falciparum enables efficient gene deletion within a single asexual erythrocytic growth cycle. Mol Microbiol. 2013; 88(4):687–701. <u>https://doi.org/10.1111/mmi.12206 PMID: 23489321</u>; PubMed Central PMCID: PMC3708112.
- Jones ML, Das S, Belda H, Collins CR, Blackman MJ, Treeck M. A versatile strategy for rapid conditional genome engineering using loxP sites in a small synthetic intron in Plasmodium falciparum. Sci Rep. 2016; 6:21800. https://doi.org/10.1038/srep21800 PMID: 26892670; PubMed Central PMCID: PMC4759600.
- 46. Thomas JA, Collins CR, Das S, Hackett F, Graindorge A, Bell D, et al. Development and Application of a Simple Plaque Assay for the Human Malaria Parasite Plasmodium falciparum. PloS one. 2016; 11(6): e0157873. https://doi.org/10.1371/journal.pone.0157873 PMID: 27332706.
- Cabrera A, Herrmann S, Warszta D, Santos JM, John Peter AT, Kono M, et al. Dissection of minimal sequence requirements for rhoptry membrane targeting in the malaria parasite. Traffic. 2012; 13 (10):1335–50. Epub 2012/07/05. https://doi.org/10.1111/j.1600-0854.2012.01394.x PMID: 22759070.
- Mueller C, Klages N, Jacot D, Santos JM, Cabrera A, Gilberger TW, et al. The Toxoplasma protein ARO mediates the apical positioning of rhoptry organelles, a prerequisite for host cell invasion. Cell Host Microbe. 2013; 13(3):289–301. Epub 2013/03/19. <u>https://doi.org/10.1016/j.chom.2013.02.001</u> PMID: 23498954.
- 49. Beck JR, Fung C, Straub KW, Coppens I, Vashisht AA, Wohlschlegel JA, et al. A Toxoplasma palmitoyl acyl transferase and the palmitoylated armadillo repeat protein TgARO govern apical rhoptry tethering

and reveal a critical role for the rhoptries in host cell invasion but not egress. PLoS Pathog. 2013; 9(2): e1003162. Epub 2013/02/15. https://doi.org/10.1371/journal.ppat.1003162 PMID: 23408890; PubMed Central PMCID: PMC3567180.

- Cao J, Kaneko O, Thongkukiatkul A, Tachibana M, Otsuki H, Gao Q, et al. Rhoptry neck protein RON2 forms a complex with microneme protein AMA1 in Plasmodium falciparum merozoites. Parasitol Int. 2009; 58(1):29–35. Epub 2008/10/28. https://doi.org/10.1016/j.parint.2008.09.005 PMID: 18952195.
- Alexander DL, Arastu-Kapur S, Dubremetz JF, Boothroyd JC. Plasmodium falciparum AMA1 binds a rhoptry neck protein homologous to TgRON4, a component of the moving junction in Toxoplasma gondii. Eukaryot Cell. 2006; 5(7):1169–73. Epub 2006/07/13. <u>https://doi.org/10.1128/EC.00040-06</u> PMID: 16835460; PubMed Central PMCID: PMC1489286.
- Ito D, Han ET, Takeo S, Thongkukiatkul A, Otsuki H, Torii M, et al. Plasmodial ortholog of Toxoplasma gondii rhoptry neck protein 3 is localized to the rhoptry body. Parasitol Int. 2011; 60(2):132–8. Epub 2011/01/18. https://doi.org/10.1016/j.parint.2011.01.001 PMID: 21237287.
- Carruthers VB, Sibley LD. Sequential protein secretion from three distinct organelles of Toxoplasma gondii accompanies invasion of human fibroblasts. Eur J Cell Biol. 1997; 73(2):114–23. Epub 1997/06/ 01. PMID: 9208224.
- Nichols BA, Chiappino ML, O'Connor GR. Secretion from the rhoptries of Toxoplasma gondii during host-cell invasion. J Ultrastruct Res. 1983; 83(1):85–98. Epub 1983/04/01. PMID: 6854716.
- Dubremetz JF, Achbarou A, Bermudes D, Joiner KA. Kinetics and pattern of organelle exocytosis during Toxoplasma gondii/host-cell interaction. Parasitol Res. 1993; 79(5):402–8. Epub 1993/01/01. PMID: 8415546.
- Sadak A, Taghy Z, Fortier B, Dubremetz JF. Characterization of a family of rhoptry proteins of Toxoplasma gondii. Mol Biochem Parasitol. 1988; 29(2–3):203–11. Epub 1988/06/01. PMID: 3045541.
- Hanssen E, Dekiwadia C, Riglar DT, Rug M, Lemgruber L, Cowman AF, et al. Electron tomography of Plasmodium falciparum merozoites reveals core cellular events that underpin erythrocyte invasion. Cell Microbiol. 2013; 15(9):1457–72. Epub 2013/03/07. https://doi.org/10.1111/cmi.12132 PMID: 23461734.
- Miller LH, Aikawa M, Johnson JG, Shiroishi T. Interaction between cytochalasin B-treated malarial parasites and erythrocytes. Attachment and junction formation. J Exp Med. 1979; 149(1):172–84. Epub 1979/01/01. https://doi.org/10.1084/jem.149.1.172 PMID: 105074; PubMed Central PMCID: PMC2184746.
- Bannister LH, Butcher GA, Dennis ED, Mitchell GH. Structure and invasive behaviour of Plasmodium knowlesi merozoites in vitro. Parasitology. 1975; 71(3):483–91. Epub 1975/12/01. PMID: 1202413.
- Aikawa M, Miller LH, Rabbege JR, Epstein N. Freeze-fracture study on the erythrocyte membrane during malarial parasite invasion. J Cell Biol. 1981; 91(1):55–62. Epub 1981/10/01. <u>https://doi.org/10.1083/</u> jcb.91.1.55 PMID: 7298726; PubMed Central PMCID: PMC2111932.
- Hakansson S, Charron AJ, Sibley LD. Toxoplasma evacuoles: a two-step process of secretion and fusion forms the parasitophorous vacuole. EMBO J. 2001; 20(12):3132–44. Epub 2001/06/19. https:// doi.org/10.1093/emboj/20.12.3132 PMID: 11406590; PubMed Central PMCID: PMC150190.
- Pinzon CG, Curtidor H, Bermudez A, Forero M, Vanegas M, Rodriguez J, et al. Studies of Plasmodium falciparum rhoptry-associated membrane antigen (RAMA) protein peptides specifically binding to human RBC. Vaccine. 2008; 26(6):853–62. Epub 2008/01/15. <u>https://doi.org/10.1016/j.vaccine.2007</u>. 11.086 PMID: 18191882.
- Oda-Yokouchi Y, Tachibana M, Iriko H, Torii M, Ishino T, Tsuboi T. Plasmodium RON12 localizes to the rhoptry body in sporozoites. Parasitol Int. 2019; 68(1):17–23. Epub 2018/10/06. https://doi.org/10.1016/ j.parint.2018.10.001 PMID: 30290224.
- 64. Chasen NM, Asady B, Lemgruber L, Vommaro RC, Kissinger JC, Coppens I, et al. A Glycosylphosphatidylinositol-Anchored Carbonic Anhydrase-Related Protein of Toxoplasma gondii Is Important for Rhoptry Biogenesis and Virulence. mSphere. 2017; 2(3). Epub 2017/05/23. https://doi.org/10.1128/ mSphere.00027-17 PMID: 28529974; PubMed Central PMCID: PMC5437132.
- 65. Collins CR, Withers-Martinez C, Hackett F, Blackman MJ. An inhibitory antibody blocks interactions between components of the malarial invasion machinery. PLoS Pathog. 2009; 5(1):e1000273. https:// doi.org/10.1371/journal.ppat.1000273 PMID: 19165323; PubMed Central PMCID: PMC2621342.
- Ling IT, Kaneko O, Narum DL, Tsuboi T, Howell S, Taylor HM, et al. Characterisation of the rhoph2 gene of Plasmodium falciparum and Plasmodium yoelii. Mol Biochem Parasitol. 2003; 127(1):47–57. Epub 2003/03/05. PMID: 12615335.
- Blackman MJ. Purification of Plasmodium falciparum merozoites for analysis of the processing of merozoite surface protein-1. Methods Cell Biol. 1994; 45:213–20. PMID: 7707987.

- **68.** Harris PK, Yeoh S, Dluzewski AR, O'Donnell RA, Withers-Martinez C, Hackett F, et al. Molecular identification of a malaria merozoite surface sheddase. PLoS Pathog. 2005; 1(3):241–51. https://doi.org/10. 1371/journal.ppat.0010029 PMID: 16322767.
- Das S, Hertrich N, Perrin AJ, Withers-Martinez C, Collins CR, Jones ML, et al. Processing of Plasmodium falciparum Merozoite Surface Protein MSP1 Activates a Spectrin-Binding Function Enabling Parasite Egress from RBCs. Cell Host Microbe. 2015; 18(4):433–44. https://doi.org/10.1016/j.chom.2015. 09.007 PMID: 26468747; PubMed Central PMCID: PMC4608996.
- 70. Thomas JA, Tan MSY, Bisson C, Borg A, Umrekar TR, Hackett F, et al. A protease cascade regulates release of the human malaria parasite Plasmodium falciparum from host red blood cells. Nat Microbiol. 2018; 3(4):447–55. Epub 2018/02/21. https://doi.org/10.1038/s41564-018-0111-0 PMID: 29459732; PubMed Central PMCID: PMC6089347.
- Kremer JR, Mastronarde DN, McIntosh JR. Computer visualization of three-dimensional image data using IMOD. J Struct Biol. 1996; 116(1):71–6. https://doi.org/10.1006/jsbi.1996.0013 PMID: 8742726.
- Zhang X, Tolzmann CA, Melcher M, Haas BJ, Gardner MJ, Smith JD, et al. Branch point identification and sequence requirements for intron splicing in Plasmodium falciparum. Eukaryot Cell. 2011; 10 (11):1422–8. Epub 2011/09/20. <u>https://doi.org/10.1128/EC.05193-11</u> PMID: <u>21926333</u>; PubMed Central PMCID: PMC3209046.
- 73. Collins CR, Hackett F, Strath M, Penzo M, Withers-Martinez C, Baker DA, et al. Malaria parasite cGMPdependent protein kinase regulates blood stage merozoite secretory organelle discharge and egress. PLoS Pathog. 2013; 9(5):e1003344. Epub 2013/05/16. https://doi.org/10.1371/journal.ppat.1003344 PPATHOGENS-D-12-03129 [pii]. PMID: 23675297; PubMed Central PMCID: PMC3649973.